Author's response to reviews

Title: Biofortification in China: Practice and Policy

Authors:

Monica H Campos (mcampos@hsc.unt.edu)
Brian F Wittenmyer (bwittenm@hsc.unt.edu)

Version: 4 Date: 25 September 2007

Author's response to reviews:

According to the last review submitted to M. Campos and B. Wittenmyer from the BioMed Central Editorial Production Team, the authors have made the following modifications to the manuscript:

Title has been modified using sentence case.

The list of authors in the manuscript has been changed to match names as they are in the submission system. All corresponding author information has been corrected.

Section headings have been changed to sentence case for all headings and sub-headings in the manuscript.

References cited in text have been changed to consecutive numbers in square brackets with corresponding numbered references at the end of the manuscript.

The reference list has been formatted in the journal’s style where applicable.

We, the authors, have made recommended revisions of this manuscript and are sending for resubmission.

Thank you,

Monica H. Campos-Bowers

Brian F. Wittenmyer